\documentclass[12pt,reqno]{article}

\usepackage[usenames]{color}
\usepackage{amssymb}
\usepackage{amsmath}
\usepackage{amsthm}
\usepackage{amsfonts}
\usepackage{amscd}
\usepackage{graphicx}

\usepackage[colorlinks=true,
linkcolor=webgreen,
filecolor=webbrown,
citecolor=webgreen]{hyperref}

\definecolor{webgreen}{rgb}{0,.5,0}
\definecolor{webbrown}{rgb}{.6,0,0}

\usepackage{color}
\usepackage{fullpage}
\usepackage{float}

\usepackage{psfig}
\usepackage{graphics}
\usepackage{latexsym}
\usepackage{epsf}
\usepackage{breakurl}

\setlength{\textwidth}{6.5in}
\setlength{\oddsidemargin}{.1in}
\setlength{\evensidemargin}{.1in}
\setlength{\topmargin}{-.1in}
\setlength{\textheight}{8.4in}

\newcommand{\seqnum}[1]{\href{https://oeis.org/#1}{\rm \underline{#1}}}
\usepackage{tikz}
\newcommand{\defn}[1]{\emph{#1}}
\newcommand{\RR}{\mathbb{R}}
\newcommand{\QQ}{\mathbb{Q}}
\newcommand{\ZZ}{\mathbb{Z}}
\newcommand{\ch}{\mathrm{ch}}
\newcommand{\NCL}{\mathrm{NCL}}
\newcommand{\db}{\delta}
\newcommand{\ec}{\epsilon}
\newcommand{\coxh}{\mathrm{h}}
\newcommand{\coxd}{\marthrm{d}}
\newcommand{\etalchar}[1]{$^{#1}$}
\def\multiset#1#2{\ensuremath{\left(\kern-.3em\left(\genfrac{}{}{0pt}{}{#1}{#2}\right)\kern-.3em\right)}}

\begin{document}
\begin{thebibliography}{8}
\bibitem{Arm09}
D.\ Armstrong, \emph{Generalized Noncrossing Partitions and Combinatorics of
  {C}oxeter Groups},  Mem.\ Amer.\ Math.\ Soc. {\bf 202},
  American Mathematical Society, 2009.

\bibitem{Ath98}
C.\ Athanasiadis, On noncrossing and nonnesting partitions for classical
  reflection groups, \emph{Electron.\ J.\ Combin.} \textbf{5} (1998), R42.

\bibitem{GV89}
I.\ Gessel and X.\ Viennot, Determinants, paths, and plane partitions,
  preprint, 1989, \url{https://people.brandeis.edu/~gessel/homepage/papers/pp.pdf}.

\bibitem{Hop22}
S.\ Hopkins, Minuscule doppelg\"angers, the coincidental down-degree expectations property, and rowmotion, \emph{Exp.\ Math.} \textbf{31} (2022)
946--974.

\bibitem{Hum90}
J.\ Humphreys, \emph{Reflection Groups and {C}oxeter Groups}, Cambridge Stud.\
  Adv.\ Math., Vol.~29, Cambridge Univ.\ Press, 1990.

\bibitem{Pro83}
R.\ Proctor, Shifted plane partitions of trapezoidal shape, \emph{Proc.\ Amer.\
  Math.\ Soc.} \textbf{89} (1983), 553--559.

\bibitem{Rei97}
V.\ Reiner, Non-crossing partitions for classical reflection groups,
  \emph{Discrete Math.} \textbf{177} (1997), 195--222.

\bibitem{sage} The Sage Developers, \emph{{S}ageMath, the {S}age {M}athematics {S}oftware {S}ystem ({V}ersion
  9.7)}, \url{https://www.sagemath.org}, 2022.

\bibitem{Sloane}  N.\ J.\ A.\ Sloane et al., The On-Line Encyclopedia of Integer Sequences, online resource, available at \url{https://oeis.org}, 2022.

\bibitem{Ste86} J.\ Stembridge, Trapezoidal chains and antichains, \emph{European.\ J.\ Combin.} \textbf{7} (1986), 377--387.

\bibitem{Ste02}
J.\ Stembridge, A weighted enumeration of maximal chains in the {B}ruhat
  order, \emph{J.\ Algebraic Combin.} \textbf{15} (2002), 291--301.
\end{thebibliography}


\begin{center}
\includegraphics[width=4in]{logo129.eps}
\end{center}

\theoremstyle{plain}
\newtheorem{theorem}{Theorem}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{proposition}[theorem]{Proposition}

\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{conjecture}[theorem]{Conjecture}

\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}

\begin{center}
\vskip 1cm{\LARGE\bf 
Nesting Nonpartitions
}
\vskip 1cm
\large
Joshua Marsh and Nathan Williams \\
Department of Mathematical Sciences\\
800 W. Campbell Road \\
Richardson, TX 75080\\ 
USA \\
\href{mailto:joshuawmarsh@gmail.com}{\tt joshuawmarsh@gmail.com} \\
\href{mailto:nathan.f.williams@gmail.com}{\tt nathan.williams1@utdallas.edu}
\end{center}

\vskip .2 in

\begin{abstract}
Antichains of root posets associated with simple complex Lie algebras
are a well-studied combinatorial object, famously counted by the
Coxeter-Catalan numbers. In this paper, we study chains in root posets
using standard enumerative techniques. The factorizations of the zeta
polynomials of these root posets gives rise to sequences that have
similar numerological properties as the exponents of the Weyl group.
\end{abstract}

\section{Introduction}

Call an interval $[a,b]$ \defn{$n$-integral} if $a$ and $b$ are integers $1 \leq a<b \leq n$. We say that two $n$-integral intervals are \defn{nesting} if one is contained in the other.  In this language, \defn{nonnesting partitions} may be defined as sets of distinct $n$-integral intervals with \emph{no} pair of intervals nesting.  Nonnesting partitions are a classical Catalan object---they are counted by the ubiquitous Catalan numbers, which appear as sequence \seqnum{A000108} in Sloane's {\it On-Line Encyclopedia of Integer Sequences}~\cite{Sloane}:
\begin{equation*}
    \mathrm{Cat}(n)=\prod_{i=1}^{n-1} \frac{n+i+1}{i+1}.
\end{equation*}

We may ask instead about \defn{nesting nonpartitions} on $n$ points: sets of distinct $n$-integral intervals such that \emph{every} pair of intervals nests.  For example, there are $1,2,6,20,68,$ and $232$ nesting nonpartitions on $1,2,3,4,5,$ and $6$ points.  The $20$ nesting nonpartitions on four points are illustrated in Figure \ref{fig:ex1}.


\begin{figure}[htbp]
\begin{center}
\scalebox{0.5}{
\begin{tikzpicture}[>=latex,line join=bevel,]
\node (node_13) at (536.0bp,167.5bp) [draw,draw=none] {
\scalebox{.4}{\begin{tikzpicture}
\filldraw [black,opacity=0.2] (1.5,0) circle(.5);
\filldraw [black,opacity=0.2] (1,0) circle(1);
\filldraw [black,opacity=0.2] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \node (node_14) at (641.0bp,114.5bp) [draw,draw=none] {\scalebox{.4}{\begin{tikzpicture}
\filldraw [black,opacity=0.2] (1,0) circle(1);
\filldraw [black,opacity=0.2] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \node (node_18) at (220.0bp,114.5bp) [draw,draw=none] {
\scalebox{.4}{\begin{tikzpicture}
\filldraw [black,opacity=0.2] (1.5,0) circle(.5);
\filldraw [black,opacity=0.2] (2,0) circle(1);
\filldraw [white,opacity=0] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \node (node_19) at (758.0bp,167.5bp) [draw,draw=none] {\scalebox{.4}{\begin{tikzpicture}
\filldraw [black,opacity=0.2] (.5,0) circle(.5);
\filldraw [black,opacity=0.2] (1,0) circle(1);
\filldraw [black,opacity=0.2] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \node (node_9) at (325.0bp,114.5bp) [draw,draw=none] {
\scalebox{.4}{\begin{tikzpicture}
\filldraw [black,opacity=0.2] (2,0) circle(1);
\filldraw [black,opacity=0.2] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \node (node_8) at (450.0bp,61.5bp) [draw,draw=none] {\scalebox{.4}{\begin{tikzpicture}
  \filldraw [black,opacity=0] (.5,0) circle(.5);
\filldraw [black,opacity=0] (2,0) circle(1);
\filldraw [black,opacity=0.2] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \node (node_17) at (852.0bp,114.5bp) [draw,draw=none] {\scalebox{.4}{\begin{tikzpicture}
\filldraw [black,opacity=0.2] (.5,0) circle(.5);
\filldraw [black,opacity=0.2] (1,0) circle(1);
\filldraw [white,opacity=0] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \node (node_6) at (126.0bp,167.5bp) [draw,draw=none] {\scalebox{.4}{\begin{tikzpicture}
\filldraw [black,opacity=0.2] (2.5,0) circle(.5);
\filldraw [black,opacity=0.2] (2,0) circle(1);
\filldraw [black,opacity=0.2] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \node (node_5) at (126.0bp,114.5bp) [draw,draw=none] {
\scalebox{.4}{\begin{tikzpicture}
\filldraw [black,opacity=0.2] (2.5,0) circle(.5);
\filldraw [black,opacity=0.2] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \node (node_4) at (536.0bp,114.5bp) [draw,draw=none] {
\scalebox{.4}{\begin{tikzpicture}
\filldraw [black,opacity=0.2] (1.5,0) circle(.5);
\filldraw [black,opacity=0.2] (1,0) circle(1);
\filldraw [white,opacity=0] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \node (node_3) at (442.0bp,114.5bp) [draw,draw=none] {
\scalebox{.4}{\begin{tikzpicture}
\filldraw [black,opacity=0.2] (1.5,0) circle(.5);
\filldraw [black,opacity=0.2] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \node (node_12) at (758.0bp,114.5bp) [draw,draw=none] {\scalebox{.4}{\begin{tikzpicture}
\filldraw [black,opacity=0.2] (.5,0) circle(.5);
\filldraw [black,opacity=0.2] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \node (node_1) at (325.0bp,167.5bp) [draw,draw=none] {
\scalebox{.4}{\begin{tikzpicture}
\filldraw [black,opacity=0.2] (1.5,0) circle(.5);
\filldraw [black,opacity=0.2] (2,0) circle(1);
\filldraw [black,opacity=0.2] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \node (node_2) at (758.0bp,61.5bp) [draw,draw=none] {\scalebox{.4}{\begin{tikzpicture}
  \filldraw [black,opacity=.2] (.5,0) circle(.5);
\filldraw [black,opacity=0] (2,0) circle(1);
\filldraw [black,opacity=0] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \node (node_7) at (588.0bp,61.5bp) [draw,draw=none] {\scalebox{.4}{\begin{tikzpicture}
  \filldraw [black,opacity=0] (.5,0) circle(.5);
\filldraw [black,opacity=.2] (1,0) circle(1);
\filldraw [black,opacity=0] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \node (node_10) at (384.0bp,61.5bp) [draw,draw=none] {\scalebox{.4}{\begin{tikzpicture}
  \filldraw [black,opacity=0.2] (1.5,0) circle(.5);
\filldraw [black,opacity=0] (2,0) circle(1);
\filldraw [black,opacity=0] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \node (node_11) at (417.0bp,8.5bp) [draw,draw=none] {\scalebox{.4}{\begin{tikzpicture}
  \filldraw [black,opacity=0] (.5,0) circle(.5);
\filldraw [black,opacity=0] (2,0) circle(1);
\filldraw [black,opacity=0] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \node (node_16) at (126.0bp,61.5bp) [draw,draw=none] {\scalebox{.4}{\begin{tikzpicture}
  \filldraw [black,opacity=0.2] (2.5,0) circle(.5);
\filldraw [black,opacity=0] (2,0) circle(1);
\filldraw [black,opacity=0] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \node (node_0) at (32.0bp,114.5bp) [draw,draw=none] {
\scalebox{.4}{\begin{tikzpicture}
\filldraw [black,opacity=0.2] (2.5,0) circle(.5);
\filldraw [black,opacity=0.2] (2,0) circle(1);
\filldraw [white,opacity=0] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \node (node_15) at (279.0bp,61.5bp) [draw,draw=none] {\scalebox{.4}{\begin{tikzpicture}
  \filldraw [black,opacity=0] (.5,0) circle(.5);
\filldraw [black,opacity=0.2] (2,0) circle(1);
\filldraw [black,opacity=0] (1.5,0) circle(1.5);
\filldraw [black,opacity=1] (0,0) circle(.15);
\filldraw [black,opacity=1] (1,0) circle(.15);
\filldraw [black,opacity=1] (2,0) circle(.15);
\filldraw [black,opacity=1] (3,0) circle(.15);
\end{tikzpicture}}};
  \draw [black,-] (node_9) ..controls (259.18bp,132.03bp) and (205.6bp,146.3bp)  .. (node_6);
  \draw [black,-] (node_11) ..controls (392.0bp,18.102bp) and (343.52bp,36.721bp)  .. (node_15);
  \draw [black,-] (node_8) ..controls (447.63bp,77.198bp) and (446.13bp,87.155bp)  .. (node_3);
  \draw [black,-] (node_11) ..controls (427.01bp,24.584bp) and (433.67bp,35.27bp)  .. (node_8);
  \draw [black,-] (node_7) ..controls (651.34bp,73.991bp) and (731.71bp,89.907bp)  .. (node_17);
  \draw [black,-] (node_11) ..controls (406.99bp,24.584bp) and (400.33bp,35.27bp)  .. (node_10);
  \draw [black,-] (node_10) ..controls (347.06bp,73.436bp) and (292.76bp,90.985bp)  .. (node_18);
  \draw [black,-] (node_16) ..controls (98.728bp,76.877bp) and (74.425bp,90.58bp)  .. (node_0);
  \draw [black,-] (node_11) ..controls (446.44bp,17.623bp) and (513.24bp,38.328bp)  .. (node_7);
  \draw [black,-] (node_9) ..controls (325.0bp,130.2bp) and (325.0bp,140.15bp)  .. (node_1);
  \draw [black,-] (node_2) ..controls (758.0bp,77.198bp) and (758.0bp,87.155bp)  .. (node_12);
  \draw [black,-] (node_15) ..controls (260.8bp,77.85bp) and (247.66bp,89.656bp)  .. (node_18);
  \draw [black,-] (node_8) ..controls (409.63bp,78.618bp) and (378.04bp,92.009bp)  .. (node_9);
  \draw [black,-] (node_3) ..controls (471.82bp,131.31bp) and (494.42bp,144.06bp)  .. (node_13);
  \draw [black,-] (node_11) ..controls (464.62bp,15.901bp) and (666.49bp,47.276bp)  .. (node_2);
  \draw [black,-] (node_15) ..controls (212.99bp,75.665bp) and (126.36bp,94.254bp)  .. (node_0);
  \draw [black,-] (node_8) ..controls (411.61bp,68.998bp) and (408.76bp,69.515bp)  .. (406.0bp,70.0bp) .. controls (308.54bp,87.137bp) and (281.54bp,89.758bp)  .. (node_5);
  \draw [black,-] (node_17) ..controls (822.18bp,131.31bp) and (799.58bp,144.06bp)  .. (node_19);
  \draw [black,-] (node_5) ..controls (126.0bp,130.2bp) and (126.0bp,140.15bp)  .. (node_6);
  \draw [black,-] (node_10) ..controls (401.81bp,77.773bp) and (414.56bp,89.425bp)  .. (node_3);
  \draw [black,-] (node_3) ..controls (404.38bp,131.54bp) and (375.18bp,144.77bp)  .. (node_1);
  \draw [black,-] (node_2) ..controls (785.27bp,76.877bp) and (809.58bp,90.58bp)  .. (node_17);
  \draw [black,-] (node_7) ..controls (571.84bp,77.97bp) and (560.62bp,89.405bp)  .. (node_4);
  \draw [black,-] (node_8) ..controls (539.33bp,76.872bp) and (641.38bp,94.433bp)  .. (node_12);
  \draw [black,-] (node_15) ..controls (293.16bp,77.815bp) and (302.82bp,88.949bp)  .. (node_9);
  \draw [black,-] (node_12) ..controls (758.0bp,130.2bp) and (758.0bp,140.15bp)  .. (node_19);
  \draw [black,-] (node_10) ..controls (399.62bp,67.629bp) and (402.92bp,68.879bp)  .. (406.0bp,70.0bp) .. controls (437.8bp,81.581bp) and (474.28bp,93.994bp)  .. (node_4);
  \draw [black,-] (node_11) ..controls (373.47bp,16.428bp) and (209.46bp,46.299bp)  .. (node_16);
  \draw [black,-] (node_14) ..controls (607.39bp,131.46bp) and (581.5bp,144.53bp)  .. (node_13);
  \draw [black,-] (node_18) ..controls (253.61bp,131.46bp) and (279.5bp,144.53bp)  .. (node_1);
  \draw [black,-] (node_7) ..controls (604.47bp,77.97bp) and (615.9bp,89.405bp)  .. (node_14);
  \draw [black,-] (node_16) ..controls (126.0bp,77.198bp) and (126.0bp,87.155bp)  .. (node_5);
  \draw [black,-] (node_0) ..controls (61.816bp,131.31bp) and (84.425bp,144.06bp)  .. (node_6);
  \draw [black,-] (node_8) ..controls (513.17bp,79.029bp) and (564.6bp,93.299bp)  .. (node_14);
  \draw [black,-] (node_4) ..controls (536.0bp,130.2bp) and (536.0bp,140.15bp)  .. (node_13);
  \draw [black,-] (node_14) ..controls (678.62bp,131.54bp) and (707.82bp,144.77bp)  .. (node_19);
\end{tikzpicture}
}
\end{center}
\caption{The 20 nesting nonpartitions on $n=4$ points, ordered by inclusion.}
\label{fig:ex1}
\end{figure}

It is not difficult to show that the number of nesting nonpartitions on $n+1$ points is given by 
\begin{equation}
\tag{\seqnum{A006012}}
\label{eq:a006012}
    \sum_{k=0}^n\sum_{i=0}^k \binom{n}{2i}\binom{n-i}{k-i}
    =\frac{\left(2+\sqrt{2}\right)^{n} + \left(2-\sqrt{2}\right)^{n}}{2}.
\end{equation}

Define a \defn{$k$-multinesting nonpartition} to be a set of $k$ $n$-integral intervals such that every pair of intervals nests (intervals may now appear with multiplicity).  As we show in Section \ref{sec:typea}, the number of $k$-multinesting nonpartitions is given by the compact expression
\begin{equation}
\label{eq:count_generic}
    \zeta_{A_{n}}(k)=\prod_{i=1}^{n-1} \frac{2k+i}{i}.
\end{equation}

Nesting nonpartitions on $n+1$ points are naturally interpreted as chains in the root poset of type $A_n$---we will review root systems and root posets in Section \ref{sec:background}.  This phrasing allows us to define the \defn{$k$-multinesting nonpartitions} for a general irreducible root system $\Phi$ as the multichains of length $k$ in the restriction of the root poset to the short roots, and state a root-theoretic generalization of Equation \ref{eq:count_generic}.

\begin{theorem}\label{thm:main} Let $\Phi$ be an irreducible crystallographic root system.  Then the number of $k$-multinesting nonpartitions is given by 
\[\zeta_{\Phi^+_s}(k)=\prod_{i=1}^{g-2} \frac{2k+\db_i}{\db_i},\]
for the sequence of positive integers $\db_1 \leq \db_2 \leq \cdots \leq \db_{g-2}=h-2$ given in Table \ref{fig:table}, where $h$ is the Coxeter number of $\Phi$, and $g$ is the dual Coxeter number of the dual root system $\Phi^{\vee}$.
\end{theorem}



Our proofs are case-by-case, and---although we show in Proposition \ref{prop:num} that the sequences $(\db_i)_{i=1}^{g-2}$ have connections to the numerology of the root system---we have been unable to find a representation-theoretic interpretation of these numbers. 

\section{Background}
\label{sec:background}

\subsection{Root systems}

Let $E$ be a finite-dimensional vector space with inner product $\langle \cdot \,, \cdot \rangle$ and let $\alpha, \beta \in E$. The reflection $\sigma_\alpha$ through the hyperplane perpendicular to $\alpha$ sends any vector parallel to $\alpha$ to its negative while leaving any vector perpendicular to $\alpha$ unchanged. By linearity, we see that \[\sigma_\alpha(\beta) = - \frac{\langle \alpha, \beta \rangle}{\langle \alpha, \alpha \rangle} \alpha + \beta - \frac{\langle \alpha, \beta \rangle}{\langle \alpha, \alpha \rangle}\alpha = \beta - 2 \frac{\langle \alpha, \beta \rangle}{\langle \alpha, \alpha \rangle} \alpha.\]


\begin{table}[htbp]
\[\begin{array}{|c|c|} \hline
W & \ec_1,\ec_2,\ldots,\ec_{g-1}  \\ \hline \hline
A_n & \begin{array}{cccc} 1 &2 &\ldots & n\end{array} \\ \hline
B_n & \begin{array}{cccc} 1 &3 &\ldots & 2n{-}1 \end{array}\\ \hline
C_n & \begin{array}{cccccc}1 & 3 & 5 & \ldots & 2n{-}3 & 2n{-}1 \\ & 3 & 5 &\ldots & 2n{-}3 & \end{array} \\ \hline
D_n & \begin{array}{cccccc}1 & 3 & 5 &\ldots & 2n{-}5 &2n{-}3 \\& & & n{-}1 & & \\ & 3 & 5 & \ldots & 2n{-}5 &\end{array}\\ \hline
E_6 & \begin{array}{ccccccccc} 1 & & 4 & 5 & & 7 & 8 & & 11 \\ & 3 & & 5 &  & 7 & & 9 & \\  & & & & 6 & & & &  \end{array}\\ \hline
E_7 & \begin{array}{ccccccccc} 1 & & 5 & 7 & 9 & 11 & 13 & & 17 \\ & 3 & 5 &7 & 9 & 11 & 13 & 15 & \\ & & & 7 & 9 & 11 & & & \end{array} \\ \hline
E_8 & \begin{array}{ccccccccccccccc} 1 & & & 7 & & 11 & 13 & & 17 & 19 & & 23 & & & 29 \\ & 3& 5 & 7 & 9 & 11 & 13 & 15 & 17 & 19 & 21 & 23 & 25 & 27 & \\ & & & & 9 & 11 & 13 & 15 &17 & 19 & 21 & & & & \\ & & & & & & & 15 & & & & & & & \end{array} \\ \hline
F_4 & \begin{array}{cccccc} 1 & & 5 & 7 & & 11 \\ & 3 & 5 &7 & 9 & \end{array} \\ \hline
G_2 & \begin{array}{ccc} 1 & & 5 \\ & 3 & \end{array} \\ \hline
\end{array}\]
\caption{The sequences $(\protect\ec_i)_{i=1}^{g-1}$ for the irreducible crystallographic root systems, arranged to illustrate their symmetry, where $\protect\ec_{i+1}=\protect\db_{i}+1$ for $1 \leq i \leq g-2$ and $\protect\ec_1=1$.   The top line for each root system consists of its exponents.}
\label{fig:table}
\end{table}

\begin{definition}
A (crystallographic) \defn{root system} is a finite set $\Phi \subseteq E$ of nonzero vectors spanning $E$ such that for $\alpha, \beta \in \Phi$ we have
\begin{itemize}
    \item $\operatorname{span}(\alpha) \cap \Phi = \{\alpha, -\alpha\}$,
    \item $\sigma_\alpha(\Phi) = \Phi$, and
    \item $2\frac{\langle \alpha, \beta \rangle }{\langle \alpha, \alpha \rangle}$ is an integer.
\end{itemize}
The \defn{Weyl group} $W$ associated with a root system $\Phi$ is the subgroup of $O(E)$ generated by the reflections $\sigma_\alpha$ for $\alpha \in \Phi$. 
\end{definition}

The condition that $2\frac{\langle \alpha, \beta \rangle }{\langle \alpha, \alpha \rangle}$ is an integer, is the crystallographic condition---it may be restated as ``$\sigma_\alpha(\beta)$ may be obtained from $\beta$ by adding an integer multiple of $\alpha$.'' A root system is \defn{reducible} if it is the union of two disjoint root systems, each of which spans one of a pair of orthogonal subspaces of $E$. Thus, a reducible root system can be viewed as the union of these two independent root systems. A root system which has no such decomposition is called \defn{irreducible}.


By the well-known classification of irreducible root systems~\cite{Hum90}, there can be at most two different root lengths.  If there are two different lengths, we call the corresponding sets of roots the \defn{short roots} and \defn{long roots}. A root system for which there is only one length of roots is called \defn{simply laced}. The root systems $A_n, D_n, E_6, E_7,$ and $E_8$ are simply laced, while $B_n, C_n, F_4,$ and $G_2$ are not.  


\begin{example}
\label{ex:roota}  
Let $E$ be the $n$-dimensional space of vectors in $\RR^{n+1}$ whose entries sum to zero, and let $e_i$ be the standard basis vectors of $\RR^{n+1}$. Define $\alpha_i = e_i - e_{i+1}$, so that $\Delta=\{ \alpha_1,\alpha_2,\ldots,\alpha_n\}$ is a basis for $E$. This set generates (by reflecting in these roots) the full root system of type $A_n$: \[\Phi_{A_n} = \{e_i-e_j : 1 \leq i,j \leq n+1\}.\]  Reflections in the hyperplane orthogonal to $e_i-e_j$ exchange the $i$th and $j$th coordinates, so that the Weyl group associated with the $A_n$ root system is isomorphic to the symmetric group $S_{n+1}$ (acting on $E$ by permuting coordinates).
\end{example}

The root system $A_n$ is often the first example of a root system, and combinatorics done in this context is often referred to as ``Type $A$,'' in contrast to combinatorics done in the context of general root systems.  More precisely, many classical combinatorial objects can be interpreted as arising from the symmetric group or $\Phi_{A_n}$ in some way, and---once phrased in that language---can often be generalized by passing to other root systems.  In this spirit, this paper interprets nesting nonpartitions in the context of type $A$ root systems and extends their definition to other root systems.

\subsection{Root posets}

By choosing a generic hyperplane in $E$ that does not contain any root, we can divide a root system $\Phi$ into positive and negative roots. The roots closest to the hyperplane form a basis for $E$, which allows us to partially order the positive roots. 

\begin{definition}
A \defn{simple system} is a subset $\Delta = \{\alpha_1, \alpha_2, \dots, \alpha_n\} \subseteq \Phi$ such that

\begin{itemize}
    \item the elements of $\Delta$ are linearly independent, and
    \item every element of $\Phi$ can be written as a linear combination of elements of $\Delta$, either with all nonnegative or all nonpositive coefficients.
\end{itemize}

\end{definition}

\begin{definition}
A set of \defn{positive roots} is a choice of roots $\Phi^+ \subseteq \Phi$ such that
\begin{itemize}
    \item for $\alpha \in \Phi$, exactly one of $\alpha$ and $-\alpha$ is in $\Phi^+$, and
    \item if a root is the sum of two roots in $\Phi^+$, it is also in $\Phi^+$. 
\end{itemize}
\end{definition}

We call the roots in the simple system the \defn{simple roots}.
The notions of positive roots and simple systems are closely related. It is easy to see that any choice of a simple system determines a set of positive roots; the roots which can be obtained from a nonnegative linear combination of simple roots may be designated as positive roots, and the rest negative. Conversely, any choice of positive roots uniquely determines a simple system~\cite[Section 1.3]{Hum90}. In particular, this means that a simple system exists, since we may choose a total order on $E$ that is compatible with the vector space operations (for example, the lexicographic order), and this order determines a positive set and therefore a simple system.  Moreover, all simple systems are essentially the same, differing only by the action of the Weyl group~\cite[Section 1.4]{Hum90}. More precisely, for any two simple systems $\Delta$ and $\Delta'$, there is some $w \in W,$ the Weyl group of the root system, so that $\Delta' = w \Delta$. Since $W \subseteq O(E)$, the simple systems are orthogonal transformations of each other and have the same geometry.  Thus, this choice is immaterial and we will make convenient choices of simple systems.

The choice of positive roots, along with vector space operations, allows us to define a partial order on the positive roots. 

\begin{definition}
\label{def:short_root}
The \defn{root poset} of an irreducible crystallographic root system $\Phi$ is the poset on the positive roots $\Phi^+$ defined by $\alpha\leq \beta$ if $\beta-\alpha$ is a nonnegative sum of positive roots.  The \defn{short root poset} $\Phi^+_s$ is the restriction of $\Phi^+$ to the short roots when there are two different root lengths, and the root poset $\Phi^+$ otherwise.
\end{definition}

\begin{example}
Continuing Example \ref{ex:roota}, the set $\Delta = \{\alpha_1, \alpha_2, \dots, \alpha_n\}$ is a convenient choice of simple roots (and will be assumed to be the set of simple roots for $A_n$ for the rest of the paper).  Writing $\alpha_{i,j} = e_i - e_j$, the corresponding positive roots are \[\Phi^+_{A_n} = \{ \alpha_{i,j} \mid 1\leq i < j \leq n+1 \}.\]    The difference between two positive roots,  $\alpha_{i,j} - \alpha_{k, l}$, is itself a nonnegative sum of simple roots when $ i \leq  k < l \leq j$.  
The $A_5$ root poset is illustrated in Figure \ref{fig:typea}.
\end{example}


\begin{figure}[htbp]
        \begin{center}
        \includegraphics{rootposeta.png} \hspace{1in}
         \includegraphics{rootposetd.png}
        \end{center}
        \caption{{\it Left}: the $A_5$ positive root poset.  {\it Right}: the $D_6$ positive root poset.}
        \label{fig:typea}
\end{figure}


\subsection{Numerology}

Associated with a root system $\Phi$ are two important invariants, the sequence of \defn{degrees} $(d_i)_{i=1}^n$ and the sequence of \defn{exponents} $(e_i)_{i=1}^n$.  It turns out that we can easily define both using the positive root poset (although this is not how they were defined historically).  A positive root $\alpha$ can be expressed as a nonnegative sum of simple roots: $\alpha=\sum_{i=1}^n a_i \alpha_i$.  Define the \defn{height} of a positive root $\alpha$ to be $\sum_{i=1}^n a_i$---the number of simple roots that must be added to get $\alpha$ (this is the rank of $\alpha$ in the positive root poset).

If $k_i$ is the number of positive roots of height $i$, we always have \[k_1 \geq k_2 \geq \dots \geq k_{h-1} = 1.\] This defines a partition of $|\Phi|/2$, the number of positive roots. From this partition, we define the sequence of exponents (in descending order) by the \emph{dual} partition $e_n \geq e_{n-1} \geq \dots \geq e_1$ of $|\Phi^+|$ with $k_1 = n$ parts.  We define the \defn{Coxeter number} by $h=e_n+1=|\Phi| / n$, the degrees by $d_i=e_i+1$, and the \defn{dual Coxeter number of the dual root system  $g$} as one plus the height of the highest short root.  In fact, it turns out that these numbers satisfy
\begin{equation}
    \label{eq:exponents}
\sum_{i=1}^{n} e_i = nh/2 \hfill \text{ and }\hfill 
	e_i+e_{n+1-i} = h.
\end{equation}

As explained in~\cite[Chapter 3]{Hum90}, there are two other algebraic settings in which the exponents and degrees appear.

The first comes from the invariant theory of the Weyl group.  Let $\Phi$ be an irreducible crystallographic root system spanning an $n$-dimensional vector space $E$, and let $W \subseteq O(E)$ be its Weyl group, generated by the reflections of $\Phi$. We associate $E$ with $\RR^n$, and $W$ acts on the polynomial ring $S = \RR[x_1,x_2, \dots, x_n]$ in a natural way by transforming the vectors $(x_1,x_2, \dots, x_n)$. Let $R$ be the ring of polynomials invariant under the action of $W$ and $f_1,f_2, \dots, f_n$ be a set of homogeneous, algebraically independent polynomials that generate $R$. Then the sequence $d_1 \leq d_2 \leq \cdots \leq d_n$, where $d_i$ is the degree of $f_i$, is independent of the choice of generating polynomials (up to reordering).  These degrees are therefore an important invariant of the root system.

The second comes from the eigenvalues of certain elements in the Weyl group.  Since any $w \in W$ is an orthogonal transformation of finite order, its eigenvalues must be roots of unity. Fix a choice of simple roots $\Delta = \{\alpha_1,\alpha_2, \dots, \alpha_n\}$ with corresponding reflections $\sigma_1,\sigma_2, \dots, \sigma_n$. Any element that is the product of all simple reflections (in any order) is called a \defn{Coxeter element}, and all such elements are conjugate, with the same order $h$.  The eigenvalues of a Coxeter element are powers of $\zeta$, where $\zeta$ is a primitive $h$th root of unity. If $(\zeta^{e_i})_{i=1}^n$ are the eigenvalues of a Coxeter element, the sequence $e_1 \leq e_2 \leq \dots \leq e_n$ recovers the exponents of $W$.

\subsection{Posets}
Recall that a \defn{chain} in a poset $P$ is a sequence of elements $p_1<p_2<\cdots<p_k$.  A \defn{multichain} is a chain with repetitions allowed.   For a finite poset $(P,\leq)$ of height $n$, write $\ch_k(P)=\{p_1\leq p_2 \leq \cdots \leq p_k\}$ for the set of multichains in $P$ of length $k$, and $\ch(P)$ for the set of maximal strict chains in $P$.   The \defn{zeta polynomial} $\zeta_P(q)$ of a finite poset $P$ is characterized as the unique polynomial satisfying \[\zeta_P(q)=|\ch_{q-1}(P)|,\] and it has leading coefficient $\frac{|\ch(P)|}{n!}$.   Write $\multiset{n}{k} = \binom{n+k-1}{k}$ for the number of ways to choose $k$ elements from a set of size $n$, with repetitions allowed.  If $a_k$ is the number of strict chains in $P$ with $k$ elements, then 
\begin{equation}
\label{eq:zeta}
\zeta_P(q) = \sum_{k=0}^\ell a_k \binom {q-2}{k-1},
\end{equation}
since there are 
\[\multiset{k}{q-1-k} = \binom{q-2}{k-1}\]
ways to choose $q-1$ elements from a set of size $k$ with repetitions allowed, subject to the requirement that every element must be selected at least once.



\section{Nesting nonpartitions}

\subsection{Nonnesting partitions}
Recall from the introduction that an interval $[a,b]$ is \defn{$n$-integral} if $a$ and $b$ are integers $1 \leq a<b \leq n$, and two $n$-integral intervals are called nesting if one is contained in the other. In this language, the \defn{nonnesting partitions} are sets of $n$-integral intervals with no pair of intervals nesting.

Associate a positive root $\alpha_{i,j}$ in the $A_n$ root system with the interval $[i, j]$, so that the order structure on positive roots is equivalent to ordering intervals by inclusion.  Then the nonnesting partitions may be rephrased as antichains in the $A_{n-1}$ root poset, giving a natural generalization to other root systems $\Phi$.

\begin{definition}
The \defn{nonnesting partitions} associated with an irreducible crystallographic root system $\Phi$ are the antichains in the positive root poset $\Phi^+$.
\end{definition}

The nonnesting partitions turn out to be counted by a root-theoretic generalization of the Catalan numbers: 
\[\mathrm{Cat}(\Phi)=\prod_{i=1}^{n-1} \frac{h + d_i}{d_i},\]
where the $(d_i)_{i=1}^n$ are the \defn{degrees} of $\Phi$, and $h=d_n$ is the Coxeter number \cite{Rei97,Ath98}.  In type $A_n$, we obtain the sequence \seqnum{A000108}, in type $B_n$ and $C_n$ we get the central binomial coefficients \seqnum{A000984}, in type $D_n$ the sequence \seqnum{A051924}, while in the exceptional types we have the numbers $\mathrm{Cat}(\Phi_{E_6})=833, \mathrm{Cat}(\Phi_{E_7})=4160, \mathrm{Cat}(\Phi_{E_8})=25080, \mathrm{Cat}(\Phi_{F_4})=105$, and $\mathrm{Cat}(\Phi_{G_2})=6$.

Nonnesting partitions have been widely studied, in part due to their connection to noncrossing partitions and cluster combinatorics.  Armstrong's monograph gives a comprehensive survey~\cite{Arm09}.

\subsection{Nesting nonpartitions}
Recall that the \defn{nesting nonpartitions} are collections of $n$-integral intervals such that \emph{every} pair of intervals nest, as illustrated in Figure \ref{fig:ex1}.  We can again interpret this definition in the context of the $A_{n-1}$ root system---the interval $[a,b]$ corresponds again to the positive root $\alpha_{a,b}$, and two intervals nest if and only if the corresponding positive roots are comparable.   Thus, it is natural to study \emph{chains} of positive roots---to our knowledge, these have not been systematically enumerated before.  

\begin{definition}
\label{def:nnon}
The \defn{nesting nonpartitions} associated with an irreducible crystallographic root system $\Phi$ are the chains in the short positive root poset $\Phi^+_s$.  The \defn{$k$-multinesting nonpartitions} are multichains with $k$ roots in $\Phi^+_s$.
\end{definition}

We emphasize that the nesting nonpartitions are counted by the zeta polynomial of the positive root poset, in contrast to the \defn{order polynomial}, which counts order ideals in $P \times [q]$.  When $P=\Phi^+$ is a positive root poset, this order polynomial counts plane partitions in $\Phi^+$; it recovers $\mathrm{Cat}(\Phi)$ for $q=1$, but has a simple product form only for roots systems of ``coincidental'' type~\cite{Hop22}.

The condition requiring chains in the \emph{short} positive root poset may seem a bit strange, but we motivate it as follows.  By Equation \ref{eq:zeta}, it suffices to study the zeta polynomial to count the objects of Definition \ref{def:nnon}.   As we saw in Equation \ref{eq:count_generic}, the zeta polynomials of the type $A$ root poset have a simple factorization; we will show the same for the other classical types $B_n,C_n$, and $D_n$.   But if we compute the zeta polynomial for the full $F_4$ positive root poset, we find it has a factor
\[q^4 + \frac{310}{39} q^3 + \frac{270}{13} q^2 + \frac{935}{39} q + \frac{142}{13},\]
which is irreducible over $\QQ$.  By restricting to the short roots, we obtain a simple factorization over $\QQ$ with desirable numerical properties, similar to the exponents.  We note that a similar phenomenon appears when considering the order polynomial of the positive root poset of type $F_4$: while this order polynomial doesn't factor into linear factors over $\QQ$, it does after restricting to the short roots.

\section{Proof of Theorem 1.1}
In this section we prove Theorem \ref{thm:main}, that the number of $k$-multinesting nonpartitions is given by 
\[\zeta_{\Phi^+_s}(k)=\prod_{i=1}^{g-2} \frac{2k+\db_i}{\db_i},\]
for the sequence of positive integers $\db_1 \leq \db_2 \leq \cdots \leq \db_{g-2}=h-2$ given in Table \ref{fig:table}, where $h$ is the Coxeter number of $\Phi$, and $g$ is the dual Coxeter number of the dual root system $\Phi^{\vee}$.  Just as the Coxeter number $h$ satisfies that $h-1$ is the height of the highest long root, the appearance of the dual Coxeter number $g$ is due to the fact that the height of the highest short root is $g-1$.

In analogy to the relationship between exponents and degrees, let $\ec_{i+1}=\db_{i}+1$ for $1 \leq i \leq g-2$ and $\ec_1=1$.  These positive integers $(\ec_i)_{i=1}^{g-1}$ are given in Figure \ref{fig:table}; these sequences contain repeated entries, and these are shown as stacked vertically in the table.   The sequence $(\ec_i)_{i=1}^{g-1}$ contains the usual exponents as a subsequence; this is shown as the top line of numbers in each entry.  For types $A$ and $B$, $(\ec_i)_{i=1}^{g-1}$ are exactly the exponents, while the other root systems contain more entries.  We have the following analogue of Equation \ref{eq:exponents}:

\begin{proposition}
\label{prop:num}
	The exponents $(e_i)_{i=1}^n$ are a subsequence of $(\ec_i)_{i=1}^{g-1}$.  Moreover,
	\[\sum_{i=1}^{g-1} \ec_i = (g-1)h/2 \text{ and } \ec_i+\ec_{g-i} = h.\]
\end{proposition}

Proposition \ref{prop:num} is easily verified by examining Figure \ref{fig:table}, and gives further evidence that the short roots are the right thing to consider. For example, in the case of $G_2$, where $h = 6$ and $g = 4$, if we do not restrict ourselves to the short roots, the sequence $(\ec_i)_{i=1}^{h-1}$ becomes $(1,3,5,5,7)$, which satisfies no such nice properties and loses the symmetry. Similarly, for types $B_n$ and $C_n$ (which have isomorphic positive root posets when not restricted to short roots), the sequence $(\ec_i)_{i=1}^{h-1}$ is $(1,3,3,5,5, \dots, 2n-1, 2n-1)$. Only types $B_n, C_n, F_4,$ and $G_2$ are not simply laced, so these are the only root systems for this modification is needed. 



\subsection{Proof in type A}
\label{sec:typea}
The Hasse diagram for the $A_n$ positive root poset is a triangle, as illustrated in Figure \ref{fig:typea} for $A_5$. The sequence $(\db_i)_{i=1}^{g-2}$ for type $A$ is given by $\db_i = i$ and $g = n+1$. The count given in Theorem \ref{thm:main} becomes \[\prod_{i=1}^{n-1} \frac{2k+i}{i} = \binom{2k+n-1}{n-1} = \multiset{n}{2k}.\]

The multichains in type $A$ are in correspondence with multisets of size $2k$ whose elements are simple roots. The $A_n$ root poset is a join semilattice, with $n$ simple roots. An arbitrary choice of two (possibly identical) simple roots can be identified with their join, and for any element of the poset, there is a unique pair of (not necessarily distinct) simple roots whose join is that element. This gives a way for $k$ not necessarily distinct elements of the poset to be projected onto a multiset of size $2k$, whose elements are all simple roots. In particular, a multichain of length $k$ can be projected in this way, starting with the smallest element and projecting one by one until the largest element has been projected. There are $\multiset{n}{2k}$ ways to multichoose $2k$ of the $n$ simple roots.

This map is bijective, which establishes the count. To go from a multiset $M$ of $2k$ simple roots to a chain, pair up the leftmost and rightmost element of $M$ and take their join to get the largest element of the chain, repeatedly pairing off the leftmost and rightmost remaining elements of $M$ to get the rest of the elements of the chain. This recovers the chain that generates the multiset, since it undoes the projection operation that generated the multiset. Furthermore, this process can be applied to any multiset $M$ of size $2k$ and must give a multichain back, showing that there is a chain that generates every multiset. Thus, there are $\multiset{n}{2k}$ multichains of length $k$ in the $A_n$ root poset.


\subsection{Proof in types B and C}

The poset of short roots in $B_n$ is just a chain of length $n$, so we see that the number of multichains of length $k$ is $\multiset{n}{k} = \binom{n+k-1}{n-1}$. This agrees with the factorization given in Theorem \ref{thm:main} with $\db_i = 2i$ and $g = n+1$.

It is convenient to prove a more general statement when considering the short roots in type $C_n$.  For $a \geq b\geq 0$, an \defn{$a\times b$ trapezoid} is the poset $T_{a,b}$ given by the integer points $(x,y)$ in $\ZZ^2$ such that $0 \leq y \leq b$ and $b-y \leq x \leq a + y$, ordered by the usual product order. As an example, the Hasse diagram for the $5 \times 2$ trapezoid is shown in Figure \ref{fig:trapezoid}.


\begin{figure}[htbp]
        \begin{center}
        \includegraphics[scale=1]{trapezoid.png}
        \end{center}
        \caption{The $5 \times 2$ trapezoid.}
        \label{fig:trapezoid}
\end{figure}

The  $C_{n}$ short root poset is an $(n-1) \times (n-2)$ trapezoid, and the number of multichains in a general $a \times b$ trapezoid is useful for the type $D$ case, so we count the number of multichains in an $a \times b$ trapezoid.  There is a terse sketch of a proof of the count given by Proctor~\cite{Pro83}; this count was also obtained by Stembridge~\cite{Ste86}. We reproduce this proof here with additional details.

The Hasse diagram of $T_{a,b}$ has $b + 1$ long diagonals, as drawn in Figure \ref{fig:trapezoid}, which we may label $0$ through $b$. Diagonal $d$ is a chain of $a+b-2d + 1$ elements. Let $E_d(m)$ denote the number of length $m$ multichains whose minimal element is on diagonal $d$. To count these, we need to know how many multichains there are in the $u \times v$  rectangle (points in $\ZZ^2$ with $0 \leq x \leq u$ and $0 \leq y \leq v$) with the elements strictly above the line $y=x$ deleted (see Figure \ref{fig:lattice-paths}). Call this poset $R_{u,v}$, with $u \geq v$, and let $F_{u,v}(m)$ be the number of length $m$ multichains in it. 

Multichains in $R_{u,v}$ are in bijection with certain pairs of non-intersecting lattice paths. Let $a_1 = (0, 1)$ and $b_1 = (v, m+1)$ be points in $\ZZ^2$. Upward steps in paths between these points correspond to the $x$-coordinate of a point in $R_{u,v}$. Let $a_2 = (1,0)$ and $b_2 = (u+1, m)$, so steps here correspond to $y$-coordinates. Specifically, if $P_1$ and $P_2$ are a pair of non-intersecting paths (no shared vertices) from $a_1$ to $b_1$ and $a_2$ to $b_2$ respectively, we may construct a multichain in $R_{u,v}$ by considering the the first ``up'' step in each path as specifying the $x$ and $y$ coordinates of the first element of the multichain, with the $x$ or $y$ coordinates given by how many total ``right'' steps there had been before that ``up'' step in the respective paths (see Figure \ref{fig:lattice-paths}). We see that these coordinates are legal so long as $P_1$ stays above $P_2$, as this ensures that the $x$-coordinate is always larger than the $y$-coordinate. We may replace $a_1$ by the point $(0,0)$ and get the same count, since the non-intersecting property requires that the first step be north, and we recover the previous situation. This modification simplifies a sum later. 

\begin{figure}[htbp] 
    \begin{center}
        \includegraphics[scale=2]{cutoffrectangle.png} \hspace{1in}
        \includegraphics[scale=1]{nclatticepaths.png}
    \end{center}
\caption{{\it Left}: A chain in $R_{4,2}$, where northeast is `up'. {\it Right}: The corresponding lattice paths.}
\label{fig:lattice-paths}
\end{figure}

This allows us to use the Lindstr\"om-Gessel-Viennot lemma \cite{GV89}, which states that the number of tuples of non-intersecting lattice paths from a starting set to an ending set is equal to the determinant of the matrix whose $i,j$th entry is the number of lattice paths from the $i$th starting point to the $j$th ending point. Thus,
\[F_{u,v}(m) = \binom{m+u}{u}\binom{m+v+1}{v} - \binom{m+u+1}{u+1}\binom{m+v}{v-1}.\]

To find $E_d(m)$, we observe that there are $F_{a+b-d,d}(m)$ $m$-multichains whose elements are on or above diagonal $d$, and $F_{a+b-d,d-1}(m)$ multichains whose elements are above but not on diagonal $d$, so there are $E_d(m) = F_{a+b-d,d}(m) - F_{a+b-d,d-1}(m)$
multichains whose minimal element is on diagonal $d$. This gives 
\[E_d(m) = 
\binom{m+a+b-d}{a+b-d}
\binom{m+d}{d} - 
\binom{m+a+b-d+1}{a+b-d+1}
\binom{m+d-1}{d-1}.\]
Summing over $d$, we find that the sum telescopes, and the only nonzero term left is the number of length $m$ multichains in an $a \times b$ trapezoid, 
\begin{equation}
\label{eq:trap}
\binom{m + a}{a}\binom{m + b}{b}.
\end{equation}

Since the short root poset for $C_n$ is an $(n-1) \times (n-2)$ trapezoid, Theorem \ref{thm:main} follows by noting that the sequence $(\db_i)_{i=1}^{g-2}$ for type $C$ is given by $(2,2, 4, 4, \dots, 2n-4, 2n-4, 2n-2)$, and that \[\prod_{i=1}^{n-1} \frac{2k+2i}{2i} \cdot \prod_{j=1}^{n-2} \frac{2k+2j}{2j} = \binom{k+n-1}{n-1} \binom{k+n-2}{n-2}.\] 

Finally, we observe that the full $B_n$ and $C_n$ root posets are $(n-1) \times (n-1)$ trapezoids, and we can see that the factorization using the full root poset does not give sequences $(\ec_i)_{i=1}^{h-1}$ with the desired numerological properties.

\subsection{Proof in type D}

The $D_n$ root poset is similar to an $(n-2) \times (n-2)$ trapezoid, but with a doubled southeast diagonal, as shown in Figure \ref{fig:typea}. We draw one of these diagonals solid, and the other dashed. We will refer to the dashed southeast diagonal as the ``dashed diagonal'' and the corresponding solid diagonal as the ``solid diagonal.'' Any multichain cannot include elements from both the solid and dashed diagonal, and we note that if we delete one of these diagonals, the resulting poset is an $(n-2) \times (n-2)$ trapezoid. If we delete both diagonals, then we have an $(n-2) \times (n-3)$ trapezoid. The total number of multichains in the type $D$ poset is the number of chains which do not pass through solid diagonal, plus the number which do not pass through the dashed diagonal, minus the number which do not pass through either diagonal. This gives the number of length $m$ multichains in the $D_n$ poset as 
\begin{align*}
&2\binom{k+n-2}{n-2}^2 - \binom{k+n-2}{n -2}\binom{k+n - 3}{n - 3}\\
&\hspace{.5cm}= \binom{k + n - 2}{n - 2}\binom{k + n - 3}{n - 3}\left(2 \cdot \frac{k+n-2}{n-2} - 1\right)\\
&\hspace{.5cm}= \binom{k + n - 2}{n - 2}\binom{k + n - 3}{n - 3}\left( \frac{2k+n-2}{n-2}\right).
\end{align*}

This exactly matches the factorization listed in Figure \ref{fig:table}, verifying the main theorem for type $D$ and thus all of the infinite families of root systems.

\subsection{Verification in the exceptional types}
The factorizations of the zeta polynomials of the short root posets for the exceptional types $E_6,E_7,E_8,F_4$ and $G_2$ were all evaluated in SageMath~\cite{sage}.  This completes the proof of the main theorem and the numerological properties for all irreducible crystallographic root systems.

\section{Future work}
The sequences associated with the zeta polynomial factorizations for the short root posets show remarkable numerological properties and relationship to the exponents across all root systems, but we have been unable to find any representation-theoretic explanation for these properties, or for why the short root poset is a more natural object to study in the root systems which are not simply-laced.  It would be interesting to find extensions of this to other root-theoretic posets, such as minuscule posets or crystals (the preceding analysis could be interpreted as multichains in the top half of the crystal associated with the highest short root), or to weighted enumerations as in~\cite{Ste02}.

It would also be interesting to compute the zeta polynomial of the poset of nesting nonpartitions, as in Figure~\ref{fig:ex1}.  In type $A$, this zeta polynomial appears to be related to the Chebyshev polynomials, while for types $B$ and $C$ these zeta polynomials appear to be related to Legendre polynomials.



\bigskip
\hrule
\bigskip

\noindent 2020 {\it Mathematics Subject Classification}:
Primary 05A18; Secondary 06A07.

\noindent \emph{Keywords: } 
root poset, nonnesting partition, zeta polynomial, short root.

\bigskip
\hrule
\bigskip

\noindent (Concerned with sequences
\seqnum{A000108},
\seqnum{A000984},
\seqnum{A006012}, and
\seqnum{A051924}.)

\bigskip
\hrule
\bigskip

\vspace*{+.1in}
\noindent
Received August 11 2022;
revised version received  October 10 2022.
Published in {\it Journal of Integer Sequences}, October 26 2022.

\bigskip
\hrule
\bigskip

\noindent
Return to
\htmladdnormallink{Journal of Integer Sequences home page}{https://cs.uwaterloo.ca/journals/JIS/}.
\vskip .1in


\end{document}

                                                                                






